Author’s response to reviews

Title: Evidence-based care of older people with suspected cognitive impairment in general practice: protocol for the IRIS cluster randomised trial

Authors:

Joanne E McKenzie (joanne.mckenzie@monash.edu)
Simon D French (s.french@unimelb.edu.au)
Denise A O'Connor (denise.oconnor@monash.edu)
Duncan S Mortimer (duncan.mortimer@monash.edu)
Colette J Browning (colette.browning@monash.edu)
Grant M Russell (grant.russell@monash.edu)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Jill J Francis (Jill.Francis.1@city.ac.uk)
Susan Michie (s.michie@ucl.ac.uk)
Kerry Murphy (kerry.murphy@monash.edu)
Fiona Kossenas (fiona.kossenas@monash.edu)
Sally E Green (sally.green@monash.edu)

Version: 4 Date: 16 July 2013

Author’s response to reviews: see over